Author's response to reviews

Title: Maternal exposure to air pollution before and during pregnancy related to changes in newborn's cord blood lymphocyte subpopulations. The EDEN Study Cohort.

Authors:

Nour Baïz (baiz@u707.jussieu.fr)
Rémy Slama (remy.slama@ujf-grenoble.fr)
Marie-Christine Béné (marie-christine.bene@medecine.uhp-nancy.fr)
Marie-Aline Charles (marie-aline.charles@inserm.fr)
Marie-Nathalie Kolopp-Sarda (mkolopps@grip.u-nancy.fr)
Antoine Magnan (antoine.magnan@univ-nantes.fr)
Olivier Thiebaugeorges (o.thiebaugeorges@maternite.chu-nancy.fr)
Gilbert Faure (faure@medecine.uhp-nancy.fr)
Isabella Annesi-Maesano (annesi-maesano@u707.jussieu.fr)

Version: 4 Date: 14 October 2011

Author's response to reviews: see over
Title: Maternal exposure to air pollution before and during pregnancy related to changes in newborn’s cord blood lymphocyte subpopulations. The EDEN Study Cohort.

Dear Editor,

We are very pleased for having our manuscript accepted for publication in your journal. Please find below the point-by-point description of the changes made.

Point-by-point description

Major revisions

1) Authors’ contributions: Please indicate that all authors read and approved the final manuscript.
Answer: the statement “All authors read and approved the final manuscript” has been added to the “Authors’ contributions” paragraph.

2) Tables: table 3 has been divided into parts, a and b. It is important that tables are numbered in ascending numerical order: 1, and 2. Please renumber the tables and any references to them in the text. Alternatively, please merge the tables into one larger table.
Answer: We choose to separate the table 3 into too different tables: Table 3 and Table 4, and all tables have been renumbered after this change.

Minor revisions

1) Line numbers: Please remove these
Answer: Line numbers have been removed.

2) Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.
Answer: All highlighting and tracking have been removed from the manuscript.

3) Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions.
Answer: Spaces between words have been checked. And the spelling of the word “lysed” has been corrected (lyzed -> lysed).